Author’s response to reviews

Title: Prevalence of Chlamydia trachomatis and Neisseria gonorrhoeae infection in adolescents in Northern Italy: an observational school-based study.

Authors:

Alberto Matteelli (alberto.matteelli@unibs.it)
Michela Capelli (michela.capelli84@gmail.com)
Giorgia Sulis (g.sulis@unibs.it)
Giuseppe Toninelli (giuseppe.toninelli@gmail.com)
Anna CC Carvalho (a.carvalho@libero.it)
Sergio Pecorelli (sergio.pecorelli@unibs.it)
Arnarlo Caruso (arnaldo.caruso@unibs.it)
Carlo Bonfanti (carlo.bonfanti@unibs.it)
Franco Gargiulo (franco.gargiulo@spedalicivili.brescia.it)
Francesco Donato (francesco.donato@unibs.it)

Version: 3  Date: 22 January 2016

Author’s response to reviews:

Dear Editor,

Thank you very much for your comments to our revised manuscript.
As required, we have included the email addresses for all authors in the title page and a conclusion heading has been introduced where suggested.

We here by submit our reply to the reviewer’s comment, the revised manuscript in track change mode, and the STROBE checklist.

In addition, English language polishing has been done for the entire manuscript.

We look forward to receiving your feedback.

Kindest regards,

Alberto Matteelli
(Corresponding author)

Point by point response to Reviewer #1

Only a further change in the manuscript I require.

The sentence at the line 283-284 "Our results suggest limited circulation of the bacterium among the adolescent population included in the survey." must be rephrased in: "Our results suggest no circulation of the infection among the adolescent population included in the survey."
Thank you very much for your comment. The text has been modified according to your suggestion (page 14 line 283).